Author’s response to reviews

Title: Inequities in coverage of smokefree space policies within the United States

Authors:

Christopher Lowrie (lowriechris779@gmail.com)
George Thomson (george.thomson@otago.ac.nz)
Amber Pearson (apearson@msu.edu)

Version: 5 Date: 24 Apr 2017

Author’s response to reviews:

The Editor, BMC Public Health

Dear Editor,

Thank you for your comments. We have now made revisions to address these comments, point by point in the response letter below.

Editor Comments:

1. To follow the journal referencing guidelines, please number all references consecutively, in square brackets, in the order in which they are cited in the text, followed by any in tables or legends. In the reference list, please list them in numerical order, and not alphabetically.

Please ensure that the reference style is followed precisely.

Our response:

References in the text are now listed in numerical order in square brackets. The references section is now in numerical order (not alphabetical).

We hope these revisions meet your requirements.

Thank you,

Amber L. Pearson, on behalf of all authors